\newcommand{\etalchar}[1]{$^{#1}$}
\begin{thebibliography}{DCG{\etalchar{+}}89}

\bibitem[AL92]{Abrahams-Larson92}
Paul~W. Abrahams and Bruce~R. Larson.
\newblock {\em Unix for the Impatient}.
\newblock Addison-Wesley, Reading, Mass., 1992.

\bibitem[ASS85]{Abelson-Sussman-Sussman85}
Harold Abelson, Gerald~Jay Sussman, and Julie Sussman.
\newblock {\em Structure and Interpretation of Computer Programs}.
\newblock The MIT Press, Cambridge, Mass., 1985.

\bibitem[Bar91]{Barkakati91}
Nabajyoti Barkakati.
\newblock {\em Unix Desktop Guide to X/Motif}.
\newblock SAMS, Carmel, Indiana, 1991.

\bibitem[Ben82]{Bentley82}
Jon~Louis Bentley.
\newblock {\em Writing Efficient Programs}.
\newblock Software Series. Prentice-Hall, Englewood Cliffs, N.J., 1982.

\bibitem[Ben86a]{Bentley86}
Jon Bentley.
\newblock Programming pearls: Little languages.
\newblock {\em Communications of the ACM}, 29(8):711--721, August 1986.

\bibitem[Ben86b]{Bentley86b}
Jon~Louis Bentley.
\newblock {\em Programming pearls}.
\newblock Addison-Wesley, Reading, Mass., 1986.

\bibitem[Ben88]{Bentley88}
Jon~Louis Bentley.
\newblock {\em More programming pearls : confessions of a coder}.
\newblock Addison-Wesley, Reading, Mass., 1988.

\bibitem[CR91]{Cameron-Rosenblatt91}
Debra Cameron and Bill Rosenblatt, editors.
\newblock {\em Learning GNU Emacs}.
\newblock O'Reilly \& Associates, Sebastopol, CA, 1991.
\newblock ISBN 0-937175-84-6.

\bibitem[Cre93]{Crevier93}
Daniel Crevier.
\newblock {\em AI: The Tumultuous History of the Search for Aritficial
  Intelligence}.
\newblock Basic Books, 1993.

\bibitem[DCG{\etalchar{+}}89]{Denning-etal89}
Peter~J. Denning, Douglas~E Comer, David Gries, Michael~C. Mulder, Allen
  Tucker, A.~Joe Turner, and Paul~R. Young.
\newblock Computing as a discipline.
\newblock {\em Communications of the ACM}, 32(1):9--23, January 1989.

\bibitem[Dew89]{Dewdney89}
A.~K. Dewdney.
\newblock {\em The Turing Omnibus : 61 excursions in computer science}.
\newblock Computer Science Press, Rockville, MD, 1989.

\bibitem[Dew90]{Dewdney90}
A.~K. Dewdney.
\newblock {\em The magic machine : a handbook of computer sorcery}.
\newblock W.H. Freeman, New York, N.Y., 1990.

\bibitem[Dij76]{Dijkstra76}
Edsger~W. Dijkstra.
\newblock {\em A Discipline of Programming}.
\newblock Prentice-Hall, Inc., Englewood Cliffs, N.J., 1976.

\bibitem[Dyb87]{Dybvig87}
R.~Kent Dybvig.
\newblock {\em The Scheme Programming Language}.
\newblock Prentice-Hall, Inc., Englewood Cliffs, New Jersey, 1987.

\bibitem[Eis88]{Eisenberg88}
Michael Eisenberg.
\newblock {\em Programming in Scheme}.
\newblock The MIT Press, Cambridge, Mass., 1988.
\newblock Edited by Harold Abelson.

\bibitem[FF87]{Friedman-Felleisen87}
Daniel~P. Friedman and Matthias Felleisen.
\newblock {\em The Little LISPer}.
\newblock The MIT Press, trade edition edition, 1987.

\bibitem[Gar91]{Gardner91}
James Gardner.
\newblock {\em Learning Unix}.
\newblock SAMS, Carmel, Indiana, 1991.
\newblock Includes disk of MKS tools.

\bibitem[Har87]{Harel87b}
David Harel.
\newblock {\em Algorithmics: the spirit of computing}.
\newblock Addison-Wesley, Reading, Mass., 1987.

\bibitem[Hod83]{Hodges83}
Andrew Hodges.
\newblock {\em Alan Turing: the enigma}.
\newblock Burnett Books, London, 1983.

\bibitem[Hof79]{Hofstadter79}
Douglas~R. Hofstadter.
\newblock {\em G\"{o}del, Escher, Bach : an eternal golden braid}.
\newblock Basic Books, New York, N.Y., 1979.

\bibitem[Hof85]{Hofstadter85}
Douglas~R. Hofstadter.
\newblock {\em Metamagical Themas: Questing for the Essence of Mind and
  Pattern}.
\newblock Basic Books, Inc., New York, N.Y., 1985.

\bibitem[Ins88]{TexasInstruments88}
Texas Instruments.
\newblock {\em PC Scheme: Student Edition}.
\newblock The Scientific Press, Redwood City, Calif., 1988.

\bibitem[Kid81]{Kidder81}
Tracy Kidder.
\newblock {\em The soul of a new machine}.
\newblock Little, Brown, Boston, 1981.

\bibitem[KP78]{Kernighan-Plauger78}
Brian~W. Kernighan and P.~J. Plauger.
\newblock {\em The Elements of Programming Style}.
\newblock McGraw-Hill, New York, N.Y., second edition, 1978.

\bibitem[Luc89]{Lucky89}
R.~W. Lucky.
\newblock {\em Silicon dreams : information, man, and machine}.
\newblock St. Martin's Press, New York, N.Y., 1989.

\bibitem[Mol88]{Mollenhoff88}
Clark~R. Mollenhoff.
\newblock {\em Atanasoff: forgotten father of the computer}.
\newblock Iowa State University Press, 1988.

\bibitem[Pap80]{Papert80}
Seymour Papert.
\newblock {\em Mindstorms : children, computers, and powerful ideas}.
\newblock Basic Books, New York, N.Y., 1980.

\bibitem[Par90]{Parnas90}
David~Lorge Parnas.
\newblock Education for computing professionals.
\newblock {\em Computer}, 23(1):17--22, January 1990.

\bibitem[Pic90]{Pickover90}
Clifford~A. Pickover.
\newblock {\em Computers, Pattern, Chaos and Beauty}.
\newblock St. Martin's Press, 1990.

\bibitem[Pic91]{Pickover91}
Clifford~A. Pickover.
\newblock {\em Computers and the imagination: visual adventures beyond the
  edge}.
\newblock St. Martin's Press, 1991.

\bibitem[Pol57]{Polya57}
G.~Polya.
\newblock {\em How to Solve It: A New Aspect of Mathematical Method}.
\newblock Princeton Science Library. Princeton University Press, second
  edition, 1957.

\bibitem[Pou85]{Poundstone85}
William Poundstone.
\newblock {\em The recursive universe : cosmic complexity and the limits of
  scientific knowledge}.
\newblock Morrow, New York, N.Y., 1985.

\bibitem[Rob92a]{Roberts92}
Ralph Roberts.
\newblock {\em Unix Desktop Guide to Emacs}.
\newblock SAMS, Carmel, Indiana, 1992.

\bibitem[Rob92b]{Robinson92}
J.~A. Robinson.
\newblock Logic and logic programming.
\newblock {\em Communications of the ACM}, 35(2):40--65, March 1992.

\bibitem[SBA92]{Schoonover-Bowie-Arnold92}
Michael~A. Schoonover, John~S. Bowie, and William~R. Arnold.
\newblock {\em GNU Emacs: Unix Text Editing and Programming}.
\newblock Hewlett Packard Press Series. Addison-Wesley, New York, N.Y., 1992.

\bibitem[SF89]{Springer-Friedman89}
George Springer and Daniel~P. Friedman.
\newblock {\em Scheme and the Art of Programming}.
\newblock McGraw-Hill, New York, N.Y., 1989.

\bibitem[Smi74]{Smith74}
Karl~J. Smith.
\newblock {\em Introduction to Symbolic Logic}.
\newblock Contemporary Undergraduate Mathematics Series. Brooks/Cole Publishing
  Co., Monterey, Calif., 1974.

\bibitem[Smu78]{Smullyan78}
Raymond Smullyan.
\newblock {\em What is the name of this book? : The riddle of Dracula and other
  logical puzzles}.
\newblock Prentice-Hall, Englewood Cliffs, N.J., 1978.

\bibitem[Sob89]{Sobell89}
Mark~G. Sobell.
\newblock {\em A Practical Guide to the Unix System}.
\newblock Benjamin/Cummings, New York, N.Y., 2 edition, 1989.

\bibitem[Tro90]{Troy90}
Douglas Troy.
\newblock {\em UNIX Systems}.
\newblock Computing Fundamentals Series. Addison-Wesley, New York, N.Y., 1990.

\bibitem[TS86]{Todino-Strang86}
Grace Todino and John Strang.
\newblock {\em Learning the UNIX Operating Sysytem}.
\newblock O'Reilly and Associates, Inc, 1986.

\bibitem[vO90]{vonOech90}
Roger von Oech.
\newblock {\em A whack on the side of the head : how you can be more creative}.
\newblock Warner Books, New York, N.Y., revised edition, 1990.

\bibitem[Wat91]{Watson91}
Mark Watson.
\newblock {\em Common LISP Modules: Artifical Intelligence in the Era of Neural
  Networks and Chaos Theory}.
\newblock Springer-Verlag, New York, N.Y., 1991.

\bibitem[Wex78]{Wexelblat78}
Richard~L. Wexelblat, editor.
\newblock {\em ACM SIGPLAN History of Programming Languages Conference, Los,
  Angeles, California}. ACM, June 1978.

\bibitem[WH90]{Ward-Halstead90}
Stephen~A. Ward and Robert~H. Halstead.
\newblock {\em Computation structures}.
\newblock The MIT electrical engineering and computer science series. MIT
  Press, Cambridge, Mass., 1990.

\bibitem[Whi93]{White93}
Ron White.
\newblock {\em How Computers Work}.
\newblock Ziff-Davis Press, Emeryville, CA, 1993.

\end{thebibliography}

\documentstyle[twocolumn]{article}
\input{use-full-page}

\begin{document}
\bibliographystyle{alpha}

\title{Becoming a Computing Professional}
\author{Gary T. Leavens \\
Department of Computer Science, Iowa State University \\
Ames, Iowa 50011-1040 USA
leavens@cs.iastate.edu %\\
\and
Charles Wright \\
Department of Electrical Engineering and Computer Engineering, Iowa State University \\
Ames, Iowa 50011 USA
wright@isuee1.ee.iastate.edu
}
\maketitle

\begin{abstract}
Becoming a computer scientist or computer engineer is more than
passing courses: it's learning how to think for yourself
as a computing professional.
This paper suggests some ways to achieve that goal.
\end{abstract}

\section{More than Skills}

How can you become a computer scientist or computer engineer?
Certainly taking courses and practice help.
But just as there is more to life than getting a job and working,
there is more to being a computing professional than having a diploma,
even more than having the needed skills.
What more?---enjoyment and intellectual stimulation.

For us, enjoyment and intellectual stimulation go hand in hand.
We also believe that if you do not enjoy your work,
you will not perform as well as you could.
Further, if you really enjoy something, then you will do better at it.
If you enjoy it, you are more likely to continually improve your skills.

The difficulty is that how well you do is also tied to your enjoyment
of a subject.  So the dilemma in learning is how you can find an enjoyable
way to increase your performance and enjoyment.

Fortunately there are many gifted authors ready to help.
These true teachers have found ways to write about their subject
in a way that is fun and enlightening.
You read them for fun, you learn more, you perform better,
and so you enjoy the subject more, which starts the cycle all over again.
It may sound like magic or paradox, but it is common knowledge in education,
and it has worked for us.
Nevertheless, if it doesn't work for you, don't despair over it.
The only test for whether you should do this or any field is whether you
do well at it and enjoy it;
there's more than one way to learn to enjoy something.

\section{Computing Literature}

The computing section of the library is mostly in the call numbers
that start with QA76.  Some Computer Engineering literature also appears
under call numbers that start with TK7885.
These books and magazines are mostly stocked on the second floor of the
Parks library, although some current issues are in the reading room,
and some reference materials are on the first floor.
Take some time and get acquainted with what's in these sections of the library,
as we can only discuss a small sample of the available literature.
Just ignore anything that you don't care about or looks too technical.

\subsection{Interesting Reading about Computing}

Douglas Hofstadter has written two classic books with a very eclectic mix
of philosophy, art, and computer science \cite{Hofstadter79}
\cite{Hofstadter85}.
(These citations, like \cite{Hofstadter79}, are keys to the references section
at the end of this paper.  Many math and science texts and journals use
a similar system for citing references.
It's wise to look them up as you go, even if it means flipping back and forth.)
In the second book, \cite{Hofstadter85}, is an introduction to LISP
(Scheme is a LISP dialect).

A. K. Dewdney has written several popular books on computing.
Highly recommended is his {\em The Turing Omnibus\/} \cite {Dewdney89}.
%Among his other books, {\em The Planiverse\/} \cite{Dewdney84}
%is not really about computers,
%but is highly entertaining, especially if you are interested in mathematics.
His book, {\em The Magic Machine\/} \cite{Dewdney90} features lots of
``programming recreations''.

A very basic introduction to the hardware and software in a personal
computer is White's book {\em How Computers Work\/}
\cite{White93}.

Computer Engineers should not miss reading Tracy Kidder's account
of the development of a new (at the time) data general minicomputer
\cite{Kidder81}.
It is also good for giving (anyone) some idea of what it's like working on
a major project.

An excellent introductory text that emphasizes
the engineering side of computing
is \cite{Ward-Halstead90}.

For some amazing computer graphics,
check out the books of Clifford Pickover \cite{Pickover90} \cite{Pickover91}.

William Poundstone's book {\em The Recursive Universe\/}
\cite{Poundstone85}
relates simple computational processes to the limits of scientific knowledge.
If you like this book, you might search out others by him.

William Robinson's book {\em Computers, Minds, and Robots\/}
\cite{Robinson92b} is an interesting philosophical discussion about the
possibility of Artificial Intelligence.  Daniel Crevier provides a fascinating
historical discussion of the same subject in his book {\em AI\/}
\cite{Crevier93}.

A good biography about Alan Turing, one of the pioneers of computing, is
\cite{Hodges83}.
A pioneer of more local fame, J. V. Atanasoff, is profiled in a biography
by Mollenhoff \cite{Mollenhoff88}.
The history of programming languages is discussed in \cite{Wexelblat78}.

Other professors recommended the following introductory books
(in addition to the above).
Information theory is closely related to computing and is
the subject of  R. W. Lucky's {\em Silicon dreams\/} \cite{Lucky89}.
An interesting view of the use of computers in education
and some other thoughts about computers is found in Seymour Papert's book
{\em Mindstorms\/} \cite{Papert80}.
David Harel's book {\em Algorithmics\/} \cite{Harel87b} may be a bit advanced,
but gives a good overview of Computer Science.

You might also want to read books about science or some other science
in general to get some perspective.
For example, Stephen Jay Gould writes wonderfully about biology,
and Issac Asimov writes well about most sciences.

\subsection{Interesting Books about Programming in General}

Books specifically about programming may be hard for you to grasp at the
beginning.  But at least sometime next term you might want to read
Abelson and Sussman's book \cite{Abelson-Sussman-Sussman85}.
Some of the insights in Dijkstra's {\em The Humble Programmer\/}
\cite{Dijkstra72} are not too difficult too understand,
and will be of help even to fairly new students.
Eventually books such as Jon Bentley's will be of interest
\cite{Bentley82} \cite{Bentley86b} \cite{Bentley88}.

An older book that discusses problems of large-scale programming
and software engineering is Brook's classic
\cite{Brooks75}.
Kennighan and Plauger's {\em The Elements of Programming Style\/}
has some good ideas about programming style \cite{Kernighan-Plauger78}.
You may eventually be interested in more mathematical approaches
to programming and program correctness, such as Dijkstra's classic
{\em A Discipline of Programming\/} \cite{Dijkstra76}.

\subsection{Magazines and Journals on Computing}

Several magazines cater to the home computer enthusiast;
some of these, such as {\em Byte\/} even feature an occasional article
about programming (among the advertisements).
But to read about programming and other computing topics,
you should try some of the following journals:
{\em Computer} (published by the IEEE Computer Society),
{\em IEEE Software}, {\em Communications of the ACM\/}
(also known as CACM), and {\em Computing Surveys} (published by the ACM).
Sometimes {\em IEEE Transactions on Education\/} and {\em SIGCSE Bulletin\/}
(published by the ACM) have good introductory articles as well.
These are available in the library, or you can subscribe.
To subscribe, join the ACM and/or the IEEE Computer Society.
Both of these have local branches, and student rates are cheap.
See the Computer Science or Computer Engineering department office
for more information about how to join.

{\em Scientific American\/} also publishes occasional articles about computers.

\section{What is Computing as an Academic Discipline?}

You can gain some perspective on computing as a discipline,
and the similarities and differences between Computer Science and Computer
Engineering, if you listen to the debate about how to teach computing.
Denning and others discuss this in an article in which they
try to give a framework for the entire ``discipline of computing''
\cite{Denning-etal89}.
A more specialized and personal treatment is \cite{Parnas90}.
Several other articles continuing these discussions have appeared
over the last few years in CACM.

\subsection{Useful Books Related to this Class}

\subsubsection{Scheme and LISP Books}

Besides our course textbook
\cite{Springer-Friedman89}
another classic book on Scheme is
{\em Structure and Interpretation of
Computer Programs\/} \cite{Abelson-Sussman-Sussman85};
read it after this course is over if you want to understand it better,
but read it sometime.
More helpful for learning Scheme is {\em The Little LISPer\/}
\cite{Friedman-Felleisen87}.
Another introductory book with many interesting examples is
\cite{Harvey-Wright94}.

A reference on Scheme and Chez Scheme is \cite{Dybvig87}.
These books are on reserve at the library.
The PC Scheme dialect of Scheme is used in the text
\cite{Eisenberg88} and described in detail in the reference
\cite{TexasInstruments88}.

Scheme is a dialect of LISP.  Although most of the Scheme books are
on reserve in the library, none of the LISP books are on reserve
(at least not for 227).
For example,
Watson's book has a lot of interesting applications written in Common LISP
\cite{Watson91}.

Books about functional programming will also have some good ideas you
can use in Scheme.

\subsubsection{Unix Books}

There are many books about how to get along in Unix.
Among the best are the following
\cite{Troy90} \cite{Sobell89} \cite{Todino-Strang86}
and \cite{Abrahams-Larson92}.
The \cite{Abrahams-Larson92} book is very much a reference,
but it is handy once you know your way around more.
If you are running Unix on a PC, look into \cite{Gardner91}.
If you want a book about workstations like those on Project Vincent,
look into \cite{Barkakati91}.

\subsubsection{Emacs Books}

Three books specifically about emacs are \cite{Roberts92}
\cite{Cameron-Rosenblatt91} and \cite{Schoonover-Bowie-Arnold92}.

\section{Thinking and Problem Solving Books}

We highly recommend the following books for sharpening your thinking
skills.
Polya's classic {\em How to Solve It\/} has been read
by generations of programmers \cite{Polya57}.
A newer book in the same vein is {\em A whack on the side of the head\/}
\cite{vonOech90}.
Also recommended is Mason's {\em Thinking Mathematically\/};
it describes the problem solving process and gives many examples.

If you want to think more logically,
look at a logic text
(\cite{Smith74} is a good introduction),
or read some of the books by Smullyan,
such as {\em What is the Name of this Book?\/} \cite{Smullyan78}.

\bibliography{journal-abbrevs,distrib,datatypes,languages,se,old-datatypes,etc}

\end{document}
